Author's response to reviews

Title: Determinants of health literacy and health behavior: a pathway model

Authors:

Xinying Sun (xysun@bjmu.edu.cn)
Yuhui Shi (yuhuishibmu@bjmu.edu.cn)
Qingqi Zeng (zengqingqi2006@126.com)
Yanling Wang (wylbi1984@sina.com)
Weijing Du (weijingdu03@163.com)
Nanfang Wei (nanfangwei2007@sina.com)
Ruiqian Xie (xierq0031@163.com)
Chun Chang (cchangbj2004@yahoo.com.cn)

Version: 2 Date: 22 November 2012

Author's response to reviews: see over
Dear Editors,

I am submitting the revised manuscript to *BMC Public Health*. It is a research article entitled “Determinants of health literacy and health behavior: a pathway model”.

We have revised this manuscript as follows.

1. We have added context information in the Background Section of abstract.

2. In terms of ethical issues, we have stated “Peking University Institutional Review approved the study”.

3. We have revised the 'Authors' contributions' section and showed each author’s contribution.

4. In the 'Competing interests' section, we declared that we have no competing interests.

Sincerely yours,

Xinying SUN, PhD.

Department of Social Medicine and Health Education

School of Public Health

Peking University Health Science Center